Author’s response to reviews

Title: Predictors and outcomes of Mycobacterium tuberculosis bacteremia among patients with HIV and tuberculosis co-infection enrolled in the ACTG A5221 STRIDE study

Authors:

John A Crump (john.crump@duke.edu)
Xingye Wu (xwu@sdac.harvard.edu)
Michelle A Kendall (kendall@sdac.harvard.edu)
Prudence D Ive (pive@witshealth.co.za)
Johnstone J Kumwenda (jonnykumwenda@hotmail.com)
Beatriz Grinsztejn (beatriz.grinsztejn@gmail.com)
Ute Jentsch (ujentsch@cls.co.za)
Susan Swindells (sswindells@unmc.edu)

Version: 2 Date: 31 July 2014

Author’s response to reviews:

Line numbering has been added and page breaks deleted.